Reviewer #1:
Comments: Immunotherapy is a hot topic for cancer therapy. The suggestion of using fixed low-dose immunotherapy agents as a potentiator in standard neoadjuvant chemotherapy regimens for oral cancer is an interesting proposition. Further research is needed to explore the role of immunotherapy in the comprehensive treatment of advanced oral cancer.

Reply: Thank you so much for the positive feedback. We do agree that further research is needed to explore further into the proposition put forth in this paper.

Reviewer #2:
Comments: Dear authors While appreciating your efforts, it is necessary to mention that:
1. In the "Keywords" section, please delete "NACT". In addition, it is better to write "Keywords" in alphabetical order.

Reply: Thank you so much for the valuable comment. We have removed NACT from the keywords and rearranged the keywords in alphabetical order.

2. The "Abstract" of the article is written exactly the same as the "Introduction". Please correct.

Reply: Thank you so much for the valuable comment. We amended the abstract as suggested.

3. It is better to remove the title "LIMITATIONS" and the text you wrote for it, without a special title, comes in a paragraph.

Reply: Thank you so much for the valuable comment. We have removed the title "LIMITATIONS" as suggested and the rest of the content is maintained as a paragraph to follow.

4. If possible, enter one or more newer references (2022 and 2023). Good luck.

Reply: Thank you so much for the valuable comment. We have added some recent references as suggested. Ref.3,10,11